% SIMIODETemplate-TeacherORStudent.tex and simiode.cls for LaTeX2e
% Version 29 July 2015.
% 
% This class (simiode.cls) is a modification of that used by 
% the journal PRIMUS - Problems, Resources, and Issues in 
% Mathematics Undergraduate Studies. 
% 
% See http://www.tandfonline.com/loi/upri .
% 
% The class was developed by  Marc Frantz with suggestions of the 
% PRIMUS Editorial Board, especially John Thoo.
%
% This template is to be used with the  class file "simiode.cls". 
%  
% The latest version of this class and template files 
% can be found at
% 
%       <http://www.SIMIODE.org> in the Author section.
%
% A "Quick Start Guide" appears below.
%
% This file must be processed with version LaTeX2e or higher.
%
\documentclass{simiode}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% With this version of the tex file and the associated SIMIODE.cls file you can produce BOTH a Student Version and a Teacher Version of your modeling scenario or technique narrative.
%
% Basically the Student Version contains the title STUDENT VERSION along with the NAME OF SCENARIO, with no abstract, no keywords, no tags, no author identifiers, and no COMMENTS section. The Student Version contains only the STATEMENT of the modeling scenario with appropriate figures, data, instructions, and perhaps a bibliography appropriate for student resourcing, but nothing in the bibliography which would "give away" the nature of the modeling scenario, etc.
%
%The Teacher Version contains the material in the Student Version, namely the title NAME OF SCENARIO, now headed with   TEACHER VERSION, but also full author  identification, abstract, key words, tag words, and, following the STATEMENT, a section of COMMENTS which can be discussion of presentation  ideas, "solutions," techniques, etc.
%
% Authors can prepare a full Teacher Version using the notions immediately below  to create a pdf for review (but in review please eliminate all information about the author as we use a double-blind referee system. Such information will be restored when accepted and published in SIMIODE at www.simiode.org.
%
%***********   Toggle the following line to turn the teacher version on (1) or off (0). ***********
%
%
\def\TeacherEdition{0} % Toggle this line's value.
\newcommand{\teacher}[1]{\ifnum\TeacherEdition=1 #1 \fi}
%
%
% The \teacher{ } command wraps any text that will appear only in the teacher version of
% the document. See this file for examples. This has been done for you in the document below.   
% Rename the pdf after you compile each final version.
% This option was created  by Eric Sullivan, Carroll College, Helena MT.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{amsmath,amssymb}
\usepackage{color}
\usepackage{graphicx}
\usepackage{hyperref}
\usepackage{url}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% NOTE regarding double spacing %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Manuscripts should be submitted doubled spaced.  This is 
% accomplished via the \baselinestretch command below.
% However, you may obtain an idea of how your article will 
% look in final form, so that you can better plan your 
% presentation.  To see your manuscript formatted very 
% similarly to the on-line published format of  SIMIODE, 
% comment the following line out by inserting a percent sign % (%) at its beginning.

\renewcommand{\baselinestretch}{1.3}

%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% NOTE regarding macros %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Any macros defined for your paper should be contained in 
% the top matter. Likewise, any environment definitions such 
% as \newtheorem or \newenvironment should also go in the 
% top matter. 
% 
% Do not \input your macros as separate files within your 
% final version because more files make it harder for the 
% editors and users to keep track of and/or modify and 
% append your paper.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Top matter is above and 
% the beginning of the document below.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{99}

\bibitem{AghajanianBing1964} Aghajanian, G. K. and O. H. L. Bing. 1964. Persistence of lysergic acid diethylamide in the plasma of human subjects.  {\it Clinical Pharmacology and Therapeutics\/}. 5: 611-614.


\bibitem{Clinton2012} Clinton, W. Bill Clinton Didn't Inhale.  \href{http://www.hark.com/clips/hlysmfhbyc-bill-clinton-didnt-inhale}{http://www.hark.com/clips/hlysmfhbyc-bill-clinton-didnt-inhale}. 

\bibitem{Cloud2007} Cloud, J. 2007. Was Timothy Leary Right? {\it Time Magazine\/}. 19 April 2007. \href{http://www.time.com/time/magazine/article/0,9171,1612717,00.html}{http://www.time.com/time/magazine/article/0,9171,1612717,00.html}.  
\bibitem{Metzler1969} Metzler, C. M. 1969.  A Mathematical Model for the Pharmacokinetics of LSD Effect. {\it Clinical Pharmacology and Therapeutics\/}. 10(5): 737-740. 


\bibitem{MKULTRA2012} Project MKULTRA. 2012. \href{http://en.wikipedia.org/wiki/Project\_MKULTRA}{http://en.wikipedia.org/wiki/Project\_MKULTRA}. 



\bibitem{WagnerEtAl1968} Wagner, J. G., G. K. Aghajanian, and O. H. L. Bing. 1968.  Correlation of performance test scores with ``tissue concentration'' of lysergic acid diethylamide in human subjects. {\it Clinical Pharmacology and Therapeutics\/}. 9(5) : 635-638.  
 


\end{thebibliography}


\newpage
% You only need to change the NAME OF SCENARIO in the next line and delete the footnote. See above for instructions on toggling from Teacher Version to Student Version.
\title{\ifnum\TeacherEdition=1 TEACHER VERSION  \else STUDENT VERSION \fi\\LSD AND PROBLEM SOLVING}









\author{Brian Winkel\\ Director
SIMIODE\\
Chardon OH 44024 USA \\
BrianWinkel@simiode.org}


\makeStitlePDFLaTex



\begin{abstract}
We describe the use of  a two compartment model of a linear system of first order linear differential equations to model  lysergic acid diethylamide (LSD) in the body. We provide the data from the literature. We offer students the opportunity to build a two compartment model to fully analyze the administration and absorption of LSD in 8 student volunteers. We also compare the model predictions with problem-solving performance on simple arithmetic activities given to subjects over the time of distribution and absorption of the LSD.
\end{abstract}


\markboth{LSD and Problem Solving}{LSD and Problem Solving}







\section*{STATEMENT}


In the 1960's there was quite a bit of interest in the drug lysergic acid diethylamide (LSD), {\it much\/} interest, from Timothy Leary \cite{Cloud2007}, to government intelligence organizations \cite{MKULTRA2012}, to scientists \cite{WagnerEtAl1968}, to individuals, to reporters writing about the 60's scene. They say that if you can remember the 60's then you did not live the 60's, for if you could remember then you were not stoned enough on drugs and were pretty square. Well, I was square and did not even inhale \cite{Clinton2012}, but I did not have an interest in academic papers in applied mathematics or LSD research back in the 60's for I was doing my graduate work in Noetherian rings and that kept me ``clean." So I missed the papers described below and did not first come upon them until browsing in the late 70's.

One of the issues surrounding the use of LSD was what it did to mind performance---enhance, detract, alter?  Thus, there were a number of studies conducted to measure the effects of LSD on  mental processes. One  study \cite{WagnerEtAl1968}  attempted to see if there was a correlation between a subject's ability to perform simple arithmetic tasks and the concentration of LSD in the tissues of the body.  In an earlier study \cite{AghajanianBing1964} five normal male volunteer subjects, ages 21 to 25,  were administered 2 mcg (ng) of LSD per kilogram of body mass doses of d-lysergic acid diethylamide (LSD-25) intravenously over a 1.5 minute period. Blood samples were then drawn at 5, 15, 30, 60, 120, 240, and 480 minutes and these were tested for  concentration levels of LSD. ``To obtain a crude index of performance, subjects were given one of a series of equivalent tests, consisting of simple addition problems, after each blood sample was drawn.''\cite[p. 612]{AghajanianBing1964} This information is given in Table~\ref{table:LSD}.

\begin{table}

\begin{center}
\begin{tabular}{|l|l|c|c|c|c|c|c|c|}\cline{2-9}
 \multicolumn{1}{c|}{}&\multicolumn{1}{|c|}{Time (hr)} &0.0833 & 0.25 & 0.5& 1.0 & 2.0 & 4.0 & 8.0 \\ \hline  \hline 
 Subject 1 & Plasma Conc (ng/ml) &11.1 & 7.4 & 6.3 & 6.9 & 5. & 3.1 & 0.8 \\
 &Perform Score (\%)&73 & 60 & 35 & 50 & 48 & 73 & 97 \\ \hline 
  Subject 2 & Plasma Conc (ng/ml) &10.6 & 7.6 & 7. & 4.8 & 2.8 & 2.5 & 2. \\
 &Perform Score (\%)&72 & 55 & 74 & 81 & 79 & 89 & 106 \\ \hline 
  Subject 3 & Plasma Conc (ng/ml) &8.7 & 6.7 & 5.9 & 4.3 & 4.4 & ---- & 0.3
   \\
 &Perform Score (\%)&60 & 23 & 6 & 0 & 27 & 69 & 81 \\ \hline 
  Subject 4 & Plasma Conc (ng/ml) &10.9 & 8.2 & 7.9 & 6.6 & 5.3 & 3.8 & 1.2 \\
 &Perform Score (\%)&60 & 20 & 3 & 5 & 3 & 20 & 62 \\ \hline 
  Subject 5 & Plasma Conc (ng/ml) &6.4 & 6.3 & 5.1 & 4.3 & 3.4 & 1.9 & 0.7 \\
 &Perform Score (\%)&78 & 65 & 27 & 30 & 35 & 43 & 51\\ \hline 
\end{tabular}


\parbox{12.5 true cm}{\caption{\label{table:LSD}Summary of data collected \cite{AghajanianBing1964, Metzler1969} on 5 male volunteers who were administered LSD and then tested on performance (Perform Score (\%)) on simple addition questions. Both performance Score and Plasma Concentrations of LSD were recorded at 5, 15, 30, 60, 120, 240, and 480 minutes after the initial infusion of LSD.}}
\end{center}

\end{table}



\subsection*{Model Building Using Compartment Models}

The papers \cite{WagnerEtAl1968, Metzler1969} use a simple compartment model in which there are two compartments, Plasma (inner) compartment and Tissue (outer) compartment. See Figure~\ref{fig:1-10-BrowseLSDCompartment}. Neither paper gives the form of the model using differential equations. Rather, \cite{WagnerEtAl1968}  gives the closed form analytic solution for concentrations of LSD in the two compartment that one can obtain from the linear system of first order differential equations for $C_P(t)$ (ng/ml), the concentration of LSD in the plasma compartment, and  $C_T(t)$ (ng/ml), the concentration of LSD in the tissue compartment. The paper  \cite{Metzler1969}  refers to these results.

In Figure~\ref{fig:1-10-BrowseLSDCompartment} we offer a model for LSD flow between plasma and tissue compartments in the human body. Defend this model, paying careful attention to assumptions and units.



\begin{figure}[h]
\centering
\includegraphics[width=12cm]{1-10-BrowseLSDCompartment}
\parbox{13 true cm}{\caption{\label{fig:1-10-BrowseLSDCompartment} Two compartment model for LSD flow between plasma and tissue compartments in the human body where the volume (in milliliters) of the plasma ($V_P$) is 16.3\% of the volume or mass  ($M$) of the body and the volume of the tissue ($V_T$) portion of the body is 11.5\% of the volume or mass of the body.}}
\end{figure}


This figuring out of units and initial concentration is sometimes a difficult bookkeeping activity, but it is essential if we are to proceed and be good applied mathematicians. The  conversion  expression (\ref{eq:convert}) 
will help. We also point out that the  ``density of human plasma (and human tissue) is about 1 kilogram per liter.'' Thus we use, where appropriate, either mass in kilograms or volume in  liters, equivalently,  as our measure.

\begin{equation}\label{eq:convert}
V_P = \underbrace{(0.163 M)}_{\rm kg \,\,of\,\, plasma} \cdot \underbrace{(1)}_{\rm liter/kg} \cdot \underbrace{(1000)}_{\rm ml/liter} \cdot = \underbrace{(163 M)}_{\rm ml\,\, of \,\, plasma}
\end{equation}


Using the notion of ``simple change in something,'' in this case amount of LSD in each compartment we can produce the system of differential equations in (\ref{eq:LSDEQ}). We discourage them from building rate of change models of just concentration  as they can be difficult with units. The last term in the equation for $C_T'(t)$ in (\ref{eq:LSDEQ}) reflects the exponential decay of the LSD in the tissue compartment due to excretion.

\begin{eqnarray} \label{eq:LSDEQ}
V_P C_P'(t) &=& k_a V_T  C_T(t) - k_b  V_P C_P(t) - k_e V_P C_P(t) \\ \nonumber
V_T C_T'(t)&=& k_b V_P C_P(t) - k_a V_T C_T(t) \, .
\end{eqnarray}
We note that $C_P(0) = 12.2699$, for originally LSD was injected  at a concentration of 2,000 ng per kg of body mass for each subject and so we have an initial concentration in the plasma of $12.2699= \frac{2000 M}{.163 M 1000}$ ng of LSD/kg of body mass. Since no LSD is in the tissues at the start (we presume!), $C_T(0)=0$. Also we have  $V_P = (0.163 M\cdot 1000) \, {\rm ml}$ and $V_T = (0.115 M \cdot 1000)\,  {\rm ml}$ in milliliters, where  $M$ is the subjects' body mass in kilograms. For the mass devoted to the plasma in a normal body is about 16.3\% while the mass devoted to the tissues is about 11.5\%. 



We need to estimate the rate term parameters $k_a$, $k_b$, and $k_e$, each with units $1/hr$ if we are to fully understand this process. In the literature \cite{WagnerEtAl1968} the average set of concentrations for all 5 subjects at each time, respectively, is used to fit the solutions using the least square criterion in which we square the difference between the analytic solution for $C_P(t_i)$ and the actual average concentration of LSD (ng/kg) in the plasma for $t = 0.833, .25, .5, 1.0, 2.0, 4.0, 8.0$ hrs. Of course we could take individual subjects and estimate their respective parameters $k_a$, $k_b$, and $k_e$, and we have done that.

You can now use {\it Mathematica\/} or another CAS to get closed form solutions for $C_P(t)$ and $C_T(t)$ from (\ref{eq:LSDEQ}). You will obtain sums of exponentials with messy terms involving $k_a$, $k_b$, and $k_e$.  If we place the average values (for all 5 subjects) of the concentration of LSD in the plasma at each time $t = 0.8333, 0.25, 0.5, 1.0, 2.0, 4.0, 8.0$ into an array, calling it $a$,  with time as the first coordinate and this concentration of LSD in the plasma at the corresponding time as the second coordinate we have  7 (from observations) time-concentration data pairs and from these we  form the sum of square errors 
(\ref{eq:LSDSSE})  between the model $C_P(t)$ values and the observed values for concentration of LSD in the plasma. Call the latter $O_1, i = 1, 2, \ldots, 7$. 

\begin{equation}\label{eq:LSDSSE}
SSE(k_a, k_b,k_e) = \sum_{i=1}^7 (C_P(t_i) - O_i)^2
\end{equation}

You are now prepared to build your final model with all the parameters and compare the model prediction with the data.

\section*{ACTIVITY}

Build a mathematical model which will predict successfully the levels of LSD in the bloodstream and correlate these levels with the performance data on the arithmetic problems given to the subjects.  Be sure to give crisp assumptions and define all variables, units, and relationships clearly and render a connected analysis. Then summarize your results.







\end{document}
